Author's response to reviews

Title: More in hope than expectation: Women's experience and expectations of pain relief in labour: A review

Authors:

Joanne E Lally (j.e.lally@ncl.ac.uk)
Madeleine J Murtagh (m.j.murtagh@ncl.ac.uk)
Sheila Macphail (Sheila.Macphail@nuth.northy.nhs.uk)
Richard G Thomson (Richard.Thomson@ncl.ac.uk)

Version: 2 Date: 4 February 2008

Author's response to reviews:

REVIEWER: Camille Raynes-Greenhow
I am grateful to this reviewer for her comments
1. References have been checked and changes made to formatting errors
2. Subtitles within the document have been changed in the light of comments made

REVIEWER: Rosemary Mander
I am grateful to this reviewer for her comments
1. The summaries of the articles have been checked and amended
2. Reference has been made to the fact that the majority of the research projects included tended to focus on pharmacological methods of pain relief. It has also been made clear throughout the article when the pain relief referred to is pharmacological.
3. Alternative methods of pain relief has been changed to non-pharmacological methods of pain relief